Author’s response to reviews

Title: What is the Influence of Tonsillectomy on the Level of Periodontal Pathogens on the Tongue Dorsum and in Periodontal Pockets

Authors:
Valeria Diener (valeria.diener@zzm.uzh.ch)
Annietta Gay (annietta.gay@gmail.com)
Michael Soyka (michael.soyka@usz.ch)
Thomas Attin (thomas.attin@zzm.uzh.ch)
Patrick Schmidlin (patrick.schmidlin@zzm.uzh.ch)
Philipp Sahrmann (philipp.sahrmann@zzm.uzh.ch)

Version: 2 Date: 19 Mar 2018

Author’s response to reviews:

Dear Sir or Madam,

the editorial changes have been performed - hopefully for your full satisfaction!

In fact, the number of the study registry had been wrong. We have corrected this mistake as indicated by you.

In this version we did not forget to upload the figures, and we added a reference for figure 3.

As criticized, we had to review the contributions by each single author.

Best wishes from Rome,

Philipp Sahrmann